\begin{table}%T1
\caption{\label{tabcompvs}Comparaison  of  our  $v\sin i$ determinations with those from other authors (mean difference, 
standard deviation, and number of common stars).}
\small%\centering
\par
\begin{tabular}{rcl}
\hline
\hline  
$\Delta$$v\sin i$ &  $n$ & source\\
\hline  
$-0.2 \pm 1.3$ & 91 & Nordstr\"om et~al. (\cite{nor04}) \\
$-0.4 \pm 1.3$ & 58 & Valenti \& Fisher (\cite{val05})\\
$-0.6 \pm 1.9$  & 21 & Tokovinin (\cite{tok92})\\
$ 0.2 \pm 1.0$   & 10 & Benz \& Mayor (\cite{ben84})\\
$-0.5 \pm 0.7$ & 10 & Gaidos et~al. (\cite{ga00})\\
\hline             
\end{tabular}
\end{table}